Author’s response to reviews

Title: Associations between human leukocyte antigen polymorphisms and hypersensitivity to antiretroviral therapy in patients with human immunodeficiency virus: a meta-analysis

Authors:

Kun Hu (hukunpku@qq.com)
Qian Xiang (xiangqz@126.com)
Zhe Wang (727286853@qq.com)
Guang-yan Mu (ninamu@126.com)
Zhuo Zhang (zhangz1023@163.com)
Ling-yue Ma (malingyue003@126.com)
Qiu-fen Xie (xqf3401252@163.com)
Shuang Zhou (zhoushuang1067@126.com)
Xiao-dan Zhang (zhangxiaodan_1993@163.com)
yimin cui (cui.pharm@pkufh.com)

Version: 4 Date: 17 Jun 2019

Author’s response to reviews:

Chuanyi Ning, PHD

Academic Editor

BMC Infectious Diseases

RE: Manuscript “Associations between human leukocyte antigen polymorphisms and hypersensitivity to antiretroviral therapy in patients with human immunodeficiency virus: a meta-analysis”.

Dear Prof. Chuanyi Ning, PHD:

“Associations between human leukocyte antigen polymorphisms and hypersensitivity to antiretroviral therapy in patients with human immunodeficiency virus: a meta-analysis”. Thank
you very much for giving us the opportunity to revise manuscript. We have revised the
manuscript accordingly. Thank you for considering our manuscript for publication in your
esteemed journal.